Author's response to reviews

**Title:** Intra-articular adhesions of Temporomandibular Joint ---- Arthroscopic Views and Clinical Perspectives

**Authors:**

- ShanYong Zhang (zhangshanyong@126.com)
- XiuMing Li (Liu.xxmm@yahoo.com.cn)
- Chi Yang (yangchi63@hotmail.com)
- XieYi Cai (Caixieyi27@126.com)
- MinJie Chen (Chenminjie2000@126.com)
- Majd S Haddad (maid-haddad@uiowa.edu)
- Bai Yun (Yunbai68@126.com)
- ZhuoZhi Chen (chaochao25@hotmail.com)

**Version:** 4  **Date:** 29 March 2009

**Author's response to reviews:** see over
Dear editor,

Thank you very much for giving me the opportunity to revise my paper (4642761652541070) according to the reviewers’ comments. The paper has been carefully revised according to the referee's comments and suggestions, and corrected by two native English speakers (USA). I hope the paper is now suitable for publication. If any problem still existed in the revision, please let me know and I will definitely have it resolved. I am looking forward to hear from you.

Best wishes,

Yours Sincerely

ShanYong ZHANG, MD, DDS

Department of OMS

Ninth People’s Hospital

School of Stomatology

Shanghai Jiaotong University

Zhi-Zao-Ju Rd., No. 639

Shanghai 200011, P.R.China

zhangshanyong@126.com